Mathematics 153 — History of Mathematics Spring 2005

Instructor: R. Schultz Office: Surge 221 Hours: 9:30–10:30 Wednesdays and by appointment Telephone: (951) 827-6459 Electronic mail: schultz@math.ucr.edu OR reinhard.schultz@ucr.edu

Course text: Burton, A History of Mathematics, Fifth Edition Online course directory: http://math.ucr.edu/∼res/math153 (Materials in this directory are only for individual use directly related to UCR activities.) Course grades: Two midterms, each 20%, three quizzes in discussion, each 5%, homework 10%, final exam 35%. Quiz dates: Tuesdays, 4/12, 5/3, 5/31. Midterms: Friday 4/22, Wednesday, 5/18. Final: Thursday 6/9, 11:30–2:30. Discussion sections: Each Tuesday; Mr. E. Overholseer is the instructor. Section 02 — 8:10 in WAT 2240, Section 03 — 9:10 in SPR 2365.

Comments on contact information Electronic mail is probably the most efficient way to get in touch. Telephone messages might not get checked for some time, and playing telephone tag can delay communications significantly. However, the local computer network is not good at filtering out unsoliticted garbage messages, so a descriptive subject heading (for example, “Math 153”) is recommended to ensure that a message does not get lost in the middle of all the unwanted trash.